Accelerator Status Report

Run Schedule: March 11, 2010 to July 22, 2010
April 28, 2010
Injector    Energy 56.62 MeV
· During yesterday’s beam studies it was found that the voltage was not stable across each helicity state of the Pockel cell. This might the cause of the Hall A helicity correlated asymmetry issues.

· The GUN is in auto mode now. 

· The search for 5 seconds beam motion continues.
Accelerator   Linac Energy 503.25 MeV
· The accelerator has been stable.

· Zones 1L18 and 2L21 were recovered.

· Cavity 0L03-1 mismatched by 40% but did not on the RF commander screen. The mismatch caused ep ion chamber trips. The problem was fixed by replacing the control module.
· Beam restored to the BSY dump and the vacuum is still good. 

· Only do the Q measurement during day shifts for now.

· (4/20) Yesterday after the Beam Studies injector beam would not transport well. Some vacuum valves would not open (2L21A and 2L22B)

· Zones 1L18 bypassed due to cooling issues. Zone 2L21 bypassed due to beam instability. EES requests that leave the zone in filament mode after bypass, which would help EES collect data.

· (4/17)BSY vacuum VBV4C00A closed and would not open.
· RF separators tripped off once.

Hall A---Pass 1, Prex from March 11 to May 24, 2010
· Hall A radiation does rates are higher than 1R/h, which requires a watch when in access.

· Hall A target chamber vacuum is not stable. We should pay attention

· Hall A has helicity-correlated asymmetry issues. During yesterday’s beam studies it was found that the voltage was not stable across each helicity state of the Pockel cell. This might the cause of the helicity correlated asymmetry issues.

· The experts are investigating.

· Hall A Compton signal to noise ratio was getting bad, which needed some fine-tuning of beam. Yesterday it was improved by tuning the injector and Hall A line.
· Hall A Moller target current limits updated. 5uA for standard runs. 15uA for the special beam mode runs. M. Joyce will modify the Maxjuice.
· Jay did a lot of Compton optics tuning and the Compton rates are good enough now. It was found that the Pb target affect the Compton rates.

· (4/14) Hall A was back online after its target and vacuum issues. The damaged target would not be used again.
· MQA1H04A degaussed and the trim card pulled.

Hall B ---Pass 3, g9-Frost from March 11 to July 22, 2010
· No problems with Hall B beam

Hall C---QWeak from May 25 to July 22, 2010
· QWeak installation
